Author's response to reviews

Title: Nutritional supplements and infection in the elderly: Why do the findings conflict?

Authors:

Saul Sternberg (saul@psych.upenn.edu)
Seth Roberts (twoutopias@gmail.com)

Version: 2 Date: 22 November 2006

Author's response to reviews:

To Nutrition Journal:

We have made the formatting changes that you requested:

> Formatting changes requested
> -----------------------------
> Affiliations - Please link these as numbers rather than figures.
> (I assumed that you mean "letters", not "figures".)
> Table and figure placement text - Please remove this.
> Running heads - Please remove.
> -----------------------------

The other changes are all very minor. I indented paragraphs to make their organization clearer, corrected a few typographical errors, made the font sizes uniform, and inserted about four words and changed two others to reduce ambiguity.

Saul Sternberg